\documentclass[10pt]{article}
\usepackage{charter,fullpage}
\usepackage[pdfborder=false,colorlinks,bookmarksnumbered,bookmarksopen,linkcolor=black,citecolor=black,urlcolor=blue]{hyperref}
\usepackage{ifthen}
\usepackage[resetlabels]{multibib}
\usepackage[ManyBibs]{currvita}
\usepackage[top=0.8in, bottom=0.8in, left=0.5in, right=0.5in]{geometry}
\usepackage{textcomp}
\usepackage{graphicx}

\newcites{refpaper}{foo}
\newcites{unrefpaper}{foo}
\newcites{journal}{foo}
\newcites{techreport}{foo}
\newcites{submitpaper}{foo}
\newcites{classproject}{foo}

% Better for lists with 1-2 items and short descriptions
\newenvironment{sublist}{\begin{list}{}{
    \setlength{\itemsep}{0em}\setlength{\parsep}{0em}%
    \setlength{\topsep}{0em}\setlength{\parskip}{0em}%
}}{\end{list}}

% Better for lists with more than 2 items and/or long descriptions
\newenvironment{subbulletlist}{\begin{list}{\labelitemii}{
    \setlength{\topsep}{\itemsep}\setlength{\parskip}{\parsep}%
}}{\end{list}}


\begin{document}

\title{Ming-Ming Cheng Curriculum Vitae}

\bibliographystyle{unsrt}
\bibliographystylerefpaper{unsrt}
\bibliographystyleunrefpaper{unsrt}
\bibliographystylesubmitpaper{unsrt}
\bibliographystylejournal{unsrt}
\bibliographystyletechreport{unsrt}
\bibliographystyleclassproject{ieeetr}

\newlength{\oldcvlabelwidth}
\renewcommand*{\cvbibname}{}

\begin{cv}{\hspace{2.3in} Ming-Ming Cheng {\large \itshape Curriculum Vitae}}


\setlength{\cvlabelwidth}{0.1em}
\begin{minipage}{.79\textwidth}\raggedleft
    \begin{cvlist}{\hspace{-0.2in}Basic Information {\small (\href{http://mmcheng.net/}{http://mmcheng.net/})}}
\item Ming-Ming Cheng is a professor with
	\href{http://cc.nankai.edu.cn/}{College of Computer and Control Engineering (CCCE)},
    \href{http://www.nankai.edu.cn/}{Nankai University}.
    He received his PhD degree from
    \href{http://www.tsinghua.edu.cn/}{Tsinghua University} in 2012 under
    guidance of \href{http://cg.cs.tsinghua.edu.cn/prof_hu.htm}{Prof. Shi-Min Hu},
    and working closely with \href{http://www0.cs.ucl.ac.uk/staff/N.Mitra/}
    {Prof. Niloy Mitra} from \href{http://www.ucl.ac.uk/}{UCL}.
    Then he did 2 years research fellow, working with
    \href{http://www.robots.ox.ac.uk/~tvg/}{Prof. Philip Torr} in Oxford, UK.
    In 2014, he joint \href{http://www.nankai.edu.cn/}{Nankai University}
    as an associate Professor.
    He became a full professor since Dec. 2016.
    \end{cvlist}
\end{minipage}%
\hspace{.2in}
\begin{minipage}{0.15\textwidth}
    \begin{flushright}
    \includegraphics[width=\linewidth]{me}
    \end{flushright}
\end{minipage}


\begin{cvlist}{Employment \& Education}
    \item[12/2016--now] Professor, \href{http://www.nankai.edu.cn/}{Nankai University}, Tianjin China.
    \item[9/2014--12/2016] Associate Professor,
        \href{http://www.nankai.edu.cn/}{Nankai University}, Tianjin China.
    \item[9/2012--9/2014] Research Fellow, work with \href{http://www.robots.ox.ac.uk/~tvg/}
        {Prof. Philip Torr} at \href{http://www.robots.ox.ac.uk/~tvg/}{Torr Vision Group}
        and \href{http://www.ox.ac.uk/}{University of Oxford}, UK.
    \item[9/2007--7/2012] PhD, Dept. of Computer Science and Technology,
        \href{http://www.tsinghua.edu.cn/}{Tsinghua University}, China
%     \begin{sublist}
%         \item Thesis: \textit{Saliency and Similarity Detection for Image Scene Analysis}.
%             Adviser: Prof. \href{http://cg.cs.tsinghua.edu.cn/prof_hu.htm}{Shi-Min Hu}
%     \end{sublist}
    \item[9/2003-7/2007] B.S., Dept. of Computer Science and Technology,
        \href{http://www.xidian.edu.cn/}{Xidian University}, China. GPA: 90.6/100 (top $1\%$)
%   \begin{sublist}
%       \item Intern in \href{http://www.microsoft.com/asia/}{MSRA}, working with \href{http://research.microsoft.com/en-us/people/fengwu/}{Feng Wu} (7/2006--8/2006)
%   \end{sublist}
\end{cvlist}

\setlength{\cvlabelwidth}{1em}
\begin{cvlist}{Research Interests}
    \item Computer graphics, computer vision, image retrieval, saliency detection, image segmentation, etc.
\end{cvlist}


\setlength{\oldcvlabelwidth}{\cvlabelwidth}
\setlength{\cvlabelwidth}{1em}
\renewcommand*{\bibindent}{.5em}
\renewcommand*{\biblabelsep}{.5em}
\begin{cvlist}{Publications \& Patents}
    \item \textbf{Representative Papers}: (\href{http://scholar.google.com/citations?user=huWpVyEAAAAJ}
    	  {9000+ Google scholar citations})
    \small
    \nociterefpaper{fan2018SOC}
    \nociterefpaper{liu2018deep}
    \nociterefpaper{18IjcaiHiFiSkeleton}
    \nociterefpaper{18IJCAI-Emeasure}
    \nociterefpaper{shi2018crowd}
    \nociterefpaper{wang2018revisiting}
    \nociterefpaper{Li18IcraVisualOdometry}
    \nociterefpaper{zhao2018flic}
    \nociterefpaper{yang2018automatic}
    \nociterefpaper{yang2018understanding}
    \nociterefpaper{CodePAMI2017}
    \nociterefpaper{SequentialOptimzationPami17}
    \nociterefpaper{FanStructMeasureICCV17}
    \nociterefpaper{bian2017gms}
    \nociterefpaper{wei2017object}
    \nociterefpaper{lin2017code}
    \nociterefpaper{liu2017richer}
    \nociterefpaper{cheng2017intelligent}
    \nociterefpaper{hou2017deeply}
    \nociterefpaper{LiuNPAR17}
    \nociterefpaper{WangDRFI2017}
    \nociterefpaper{wei2017stc}
    \nociterefpaper{cheng2016hfs}
    \nociterefpaper{StruckPAMI}
    \nociterefpaper{cgf/Cheng15}
    \nociterefpaper{SalObjBenchmark}
    \nociterefpaper{SemanticPaint}
    \nociterefpaper{ChengPAMI}
    \nociterefpaper{14ImgSpirit}
    \nociterefpaper{BingObj2014}
    \nociterefpaper{AttObj2014}
    \nociterefpaper{ChengGroupSaliency14}
    \nociterefpaper{13iccv_Cheng_Saliency}
    \nociterefpaper{13iccv_Fitting}
    \nociterefpaper{VisualMediaSurvey13}
    \nociterefpaper{PoseShop12}
    \nociterefpaper{CMM12THU}
    \nociterefpaper{InterImgs12}
    \nociterefpaper{ImgAdmix12}
    \nociterefpaper{Saliency11}
    \nociterefpaper{pami11/ChengZ}
    \nociterefpaper{tog/ChengZMHH10}
    \nociterefpaper{tog/ChenCT09}
    \nociterefpaper{cgf/ZhangC09}
    \bibliographystyle{unsrt}
    \bibliographyrefpaper{vitae}
    \item {\small \textbf{Notes:} \textbf{IEEE TPAMI} is the \#1 IEEE publication. \textbf{ACM TOG} is the
        \href{http://dev.acm.org/publications/reviews/isiranking}{most-cited} ACM Transaction.
        \textbf{IEEE TPAMI}, \textbf{ACM TOG}, and \textbf{IEEE TVCG} are $\mathbf{A}$ class journals in the
        \href{http://www.ccf.org.cn/sites/ccf/paiming.jsp}{CCF recommendation list}.
        \textbf{SIGGRAPH}, \textbf{CVPR}, and \textbf{ICCV} have CiteSeer impact factor ranking
        top 0.7\%, top 5\%, and top 5\% of all computer science journals and conferences. }
    \item \textbf{Patents}:  \href{http://www.soopat.com/Home/Result?SearchWord=FMR\%3A(\%E7\%A8\%8B\%E6\%98\%8E\%E6\%98\%8E)\%20DZ\%3A(\%E6\%B8\%85\%E5\%8D\%8E\%20OR\%20\%E8\%A5\%BF\%E5\%AE\%89\%E7\%94\%B5\%E5\%AD\%90)\%20}
        {10 Chinese patents} granted and \href{http://worldwide.espacenet.com/searchResults?compact=false&ST=advanced&IN=\%22MingMing+Cheng\%22&locale=en_EP&DB=EPODOC&PA=Tsinghua+or+Shaofu}
        {3 International patents} are published.
\end{cvlist}

\begin{cvlist}{Major Honors \& Awards}
    \small
    \item[\hspace{.4in} 10/2016] ACM China Raising Star Award
    \item[\hspace{.48in} 8/2013] \href{http://zfxxgk.beijing.gov.cn/columns/63/5/410279.html}
        {Outstanding Doctoral Dissertations Award} from Beijing Municipal Government
    \item[\hspace{.48in} 6/2012] Beijing Municipal Government Award for ``Outstanding PhD Graduates"
    \item[\hspace{.48in} 8/2011] \href{https://www.ibm.com/developerworks/university/phdfellowship/}
        {IBM Ph.D. Fellowship}
    \item[\hspace{.4in} 12/2010] \href{http://googleblog.blogspot.com/2010/07/google-phd-fellowships-go-international.html}
        {Google PhD Fellowship}
    \item[\hspace{.6in} 2010] \href{http://baike.baidu.com/view/3628082.htm}
        {Scholarship Award for Excellent Doctoral Student} funded by Ministry of Education
    \item[\hspace{.48in} 6/2006] Microsoft \href{http://en.wikipedia.org/wiki/Windows_ChallengE}
        {Windows Embedded Student ChallengE}, 3rd Place in the World Final Competition. ($\$4,000$)
\end{cvlist}


\setlength{\cvlabelwidth}{1em}
\begin{cvlist}{Highlighted Research Experience}
    \small
    \item \textbf{Objectness estimation:} This work presented a novel method for \emph{objectness} estimation,
        which  is usually represented as a value which reflects how likely an image window covers an object
        of \emph{any category} image windows.
        We show on the standard {\sc pascal voc} benchmark that the proposed method is 1000$\times$ faster
        than competitive methods, while producing better results.
        The corresponding C++ source code got 500+ downloads in 2 days after its initial release!
        This paper has been published in \textbf{IEEE CVPR (Oral)} 2014.
    \item \textbf{Saliency detection:}  This work aims at the fundamental computer vision
        problem of salient object detection.
        Its accuracy (precision = $90\%$, recall = $90\%$) is much better than previous best
        results (precision = $75\%$, recall = $83\%$),
        when evaluated on the largest public available dataset.
        The source code has been request by and shared to more than 1000+ individuals for research or commercial usage.
        My saliency detection paper is currently \href{http://arnetminer.org/conferencebestpapers}
        {the $2^{nd}$ most cited paper} in all 400+ IEEE CVPR 2011 papers,
        and its journal version published at \textbf{IEEE TPAMI} 2015.
    \item \textbf{Sketch2Photo:} This work was selected as one of the \emph{10 most innovative and promising
        worldwide initiatives of 2009} by the \href{http://www.netexplo.org/}{Netexplorateur}
        jury, under the aegis of French government.
        The related introduction video in \href{http://vimeo.com/6496886}{Vimeo} has attracted
        $1,000,000+$ clicks within 6 months.
        It also received the best research achievement award in China National Computer Conference 2010.
        The online \href{http://innov.labs.qq.com/official/sketch2photo/}{commercial software} also draws great user attention.
        I'm in charge of the retrieval algorithm (first half) of this work.
        The corresponding research paper was published in \textbf{ACM TOG} 2009.
    \item \textbf{Project experiences:} I participated in several national
        (\href{http://www.973.gov.cn/English/Index.aspx}{973},
        \href{http://www.most.gov.cn/eng/programmes1/200610/t20061009_36225.htm}{863},
        \href{http://en.wikipedia.org/wiki/National_Natural_Science_Foundation_of_China}{NSFC}, etc.)
        and international
        (\href{http://www.epsrc.ac.uk/}{EPSRC}, etc.) research program.
        %National Basic Research Program (973 program) of China,
        %National High Technology Research and Development Program (863 Program) of China,
        %National Natural Science Foundation of China,
    %\begin{sublist} \item something \end{sublist}
\end{cvlist}
\setlength{\cvlabelwidth}{\oldcvlabelwidth}


\begin{cvlist}{Professional Activities}
    \small
    \item \textbf{Program/Orgnization Co-Chair: }
        VALSE
        \href{http://mc.eistar.net/valse2016/}{2016},
        CVM \href{http://iccvm.org/2017/}{2017},
        CCCV \href{http://prcv-conf.org/2017}{2017}
    \item \textbf{Publicity/Website Chair: }
        ACCV \href{http://www.accv2016.org/}{2016},
        CVPR \href{http://cvpr2019.thecvf.com/}{2019},
%    \item \textbf{Program committee: }
%        Pacific Graphics \href{http://cg.cs.tsinghua.edu.cn/pg2015/}{2015},
%        ICME \href{http://www.icme2014.org/}{2014},
%        ACCV \href{http://www.accv2014.org/}{2014}, \href{http://www.accv2012.org}{2012},
%        ChinaGraph \href{http://cs.nju.edu.cn/chinagraph2010/}{2010},
%        etc.
    %\begin{subbulletlist} \item \end{subbulletlist}
%    \item \textbf{Journal reviewer: }
%        IEEE
%        \href{http://ieeexplore.ieee.org/xpl/RecentIssue.jsp?punumber=34}{TPAMI},
%        \href{http://www.signalprocessingsociety.org/publications/periodicals/image-processing/}{TIP},
%        \href{http://mc.manuscriptcentral.com/tmm-ieee}{TMM},
%        \href{http://www.computer.org/portal/web/tvcg}{TVCG},
%        \href{http://ieeexplore.ieee.org/xpl/RecentIssue.jsp?punumber=76}{TCSVT}
%        \href{http://mc.manuscriptcentral.com/cyb-ieee}{CYB},
%        \href{http://mc.manuscriptcentral.com/spl-ieee}{SPL};
%        Elsevier \href{http://www.elsevier.com/locate/patrec}{PRL},
%        \href{http://www.journals.elsevier.com/image-and-vision-computing/}{IMAVIS},
%        \href{http://www.journals.elsevier.com/computers-and-graphics/}{CAG},
%        \href{http://ees.elsevier.com/image/default.asp}{Image Communication},
%        \href{http://www.journals.elsevier.com/journal-of-visual-communication-and-image-representation/}{JVCI},
%        \href{http://www.journals.elsevier.com/neurocomputing/}{NEUCOM},
%        \href{http://ees.elsevier.com/sigpro/default.asp}{SIGPRO};
%        Wiley \href{http://onlinelibrary.wiley.com/journal/10.1111/(ISSN)1467-8659}{CGF};
%        Springer \href{http://www.springer.com/computer/image+processing/journal/371}{TVC},
%        \href{http://www.editorialmanager.com/nepl/}{NEPL},
%        \href{http://mc03.manuscriptcentral.com/scis}{SCIS},
%        \href{http://www.springer.com/computer/journal/11714}{ZUSC},
%        \href{http://www.springer.com/computer/journal/11390}{JCST};
%        IET \href{http://digital-library.theiet.org/content/journals/iet-cvi}{CVI},
%        etc.
%    \item \textbf{Conference reviewer: }
%        SIGGRAPH \href{http://s2013.siggraph.org/}{2013},
%        SIGGRAPH Asia \href{http://sa2014.siggraph.org/}{2014},
%        \href{http://www.siggraph.org/asia2011/}{2011},
%        Eurographics \href{http://www.eurographics2012.it/}{2012},
%        Pacific Graphics \href{http://graphics.csie.ncku.edu.tw/PG2011/}{2011},
%        GMP \href{http://gmp.sce.ntu.edu.sg/}{2014},
%        ACCV \href{http://www.accv2014.org/}{2014},
%        \href{http://www.accv2010.org/}{2010}, etc.
%    \item \textbf{Invited talks: }
%        \href{http://www.aearu.org/}{AEARU} 2011 workshop in Kyoto Japan,
%        \href{http://cs.nju.edu.cn/chinagraph2010/}{ChinaGraph 2010} in Nanjing China,
%    \item \textbf{Oral presentations: }
%        IEEE CVPR \href{http://www.pamitc.org/cvpr14/}{2014},
%        ACM SIGGRAPH \href{https://www.siggraph.org/s2010/}{2010},
%        Pacific Graphics \href{http://cg.postech.ac.kr/pg2009/}{2009}
%    \item \textbf{Fast-forward/spotlights presentations: }
%        IEEE ICCV \href{http://www.iccv2013.org/}{2013},
%        ACM SIGGRAPH Asia \href{https://www.siggraph.org/s2010/}{2009}
%    \item \textbf{PhD students:}
%        Shuai Zheng (co-supervised with Prof. Philip Torr at Oxford University)
%    \item \textbf{Invited talks}
%        Cambridge University \href{http://talks.cam.ac.uk/talk/index/52159}{2014/4},
%        Oxford University,  \href{http://www.robots.ox.ac.uk/~vgg/}{VGG group, 2012/11},
%        Cardiff University \href{http://www.cs.cf.ac.uk/research/vis.php}{2013/1}
\end{cvlist}

%\begin{cvlist}{Experiences}
%    \small
%    \item[7/2011--7/2012] Vice-chairman of CCF \href{http://www.yocsef.org.cn}{YOCSEF} GS
%        (A national academic organization for graduated students)
%\end{cvlist}

\begin{cvlist}{Media Coverage}
    \small
    \item My works have been reported by several famous international media, including
        \href{http://www.bbc.com/news/technology-33379653} {BBC},
        \href{http://www.huffingtonpost.com/2009/10/06/photosketch-program-conve_n_310861.html} {Huffington Post},
        \href{http://www.telegraph.co.uk/technology/news/6265876/PhotoSketch-picture-software-wins-plaudits.html} {UK Telegraph},
        \href{http://www.spiegel.de/netzwelt/gadgets/bildtrickser-photosketch-diese-software-errechnet-fotos-aus-gekritzel-a-653734.html} {Spiegel},
        etc.
\end{cvlist}


\end{cv}
\end{document}

